Author’s response to reviews

Title: Prevalence and correlates of physical disability and functional limitation among community dwelling older people in rural Malaysia, a middle income country

Authors:

Noran N Hairi (noran@um.edu.my)
Awang Bulgiba (awang@um.edu.my)
Robert G Cumming (robert.cumming@sydney.edu.au)
Vasi Naganathan (vasi.naganathan@sydney.edu.au)
Izzuna M Ghazali (drizzuna@moh.gov.my)

Version: 5 Date: 17 August 2010

Author’s response to reviews:

The revisions were:
Page 1: In the title, Malaysia is spell with capital M